

# **Clusters of Disease Activity and Early Risk Factors of Clinical Course of Pediatric Crohn's Disease**

Manuela Distante, MD,<sup>\*,a,</sup> Silvia Rotulo, MD,<sup>\*,a</sup> Marco Ranalli, MD,<sup>\*,a</sup> Eugenio Pedace, MD,<sup>\*</sup> Paolo Lionetti, MD, PhD,<sup>†,</sup> Serena Arrigo, MD,<sup>‡</sup> Patrizia Alvisi, MD,<sup>§</sup> Erasmo Miele, MD,<sup>¶</sup> Massimo Martinelli, MD,<sup>¶</sup> Giovanna Zuin, MD,<sup>¶</sup> Matteo Bramuzzo, MD,<sup>\*\*,</sup> Mara Cananzi, MD,<sup>††</sup> and Marina Aloi, MD, PhD,<sup>\*,</sup> on behalf of SIGENP IBD Working Group

From the\*Pediatric Gastroenterology and Liver Unit, Department of Maternal and Child Health, Umberto I Hospital, Sapienza University of Rome, Rome, Italy

<sup>†</sup>Unit of Gastroenterology and Nutrition, Meyer Children's Hospital, Florence, Italy

<sup>‡</sup>Gastroenterologia ed Endoscopia Pediatrica, IRCCS Istituto Giannina Gaslini, Genoa, Italy

<sup>§</sup>Pediatric Gastroenterology Unit, Maggiore Hospital, Bologna, Italy

Department of Translational Medical Science, Section of Pediatrics, University of Naples Federico II, Naples, Italy

<sup>1</sup>Pediatrics, Fondazione IRCCS San Gerardo dei Tintori, Monza, Italy

\*\*Institute for Maternal and Child Health, IRCCS Burlo Garofolo, Trieste, Italy

<sup>11</sup>Unit of Gastroenterology, Digestive Endoscopy, Hepatology and Care of Children with Liver Transplantation, University Hospital of Padova, Padova, Italy

<sup>a</sup>The authors contributed equally.

Address correspondence to: Marina Aloi, MD, PhD, Department of Maternal and Child Health, Pediatric Gastroenterology and Liver Unit, Sapienza University of Rome, Viale Regina Elena 324, 00161 Roma, Italy (marina.aloi@uniroma1.it).

Background: This study aimed to define clusters of disease activity and prognostic factors of disease course in a well-characterized cohort of children with Crohn's disease (CD).

**Methods:** All patients from the SIGENP IBD (Italian Society of Pediatric Gastroenterology Hepatology and Nutrition Inflammatory Bowel Disease) registry with a 5-year follow-up and 6-monthly evaluation were included. Active disease was defined for each semester as follows: clinical activity (weighted Pediatric Crohn's Disease Activity Index  $\geq$ 12.5 or Mucosal Inflammation Noninvasive Index  $\geq$ 8) and active disease on endoscopy (Simple Endoscopic Score for Crohn's Disease >3 or fecal calprotectin >250 µg/g) or imaging. Formula-based clusters were generated based on previously published patterns in adults.

**Results:** Data from 332 patients were analyzed. A total of 105 (32%) experienced a quiescent disease course; 49 (15%) and 31 (9%) a moderateto-severe chronically active and chronic intermittent disease, respectively; 104 (31%) and 43 (13%) had active disease in the first 2 years after diagnosis and remission thereafter and vice versa, respectively. Surgery at diagnosis was significantly associated with a quiescent course (odds ratio [OR], 10.05; 95% confidence interval [CI], 3.05-25.22; P=.0005), while growth impairment at the diagnosis and active disease requiring corticosteroids at 6 months were inversely related to the quiescent group (OR, 0.48; 95% CI, 0.27-0.81; P=.007; and OR, 0.35; 95% CI, 0.16-0.71; P=.005, respectively). Perianal involvement at diagnosis and moderate–severe activity at 6 months correlated with disease progression (OR, 3.85; 95% CI, 1.20-12.85; P=.02).

**Conclusions:** During the first 5 years of follow-up, one-third of children with CD experience a quiescent course. However, another one-third have a moderate-to-severe disease course. Surgery at the diagnosis is related to a quiescent course, while growth impairment and lack of response to induction therapy correlate with more severe disease activity during follow-up.

## Lay Summary

We aimed to define clusters of disease activity and prognostic factors of disease course in pediatric Crohn's disease. One-third of patients have a quiescent course; however, half of them have an active disease by the end of the 5-year follow-up.

Key Words: cluster, Crohn's disease, prognostic risk factors, children

# Introduction

Pediatric Crohn's disease (CD) is a complex condition to diagnose and treat due to its heterogeneous clinical manifestations. Children with CD often present with extensive intestinal involvement and complicated disease behaviors, requiring early use of immunomodulators and/or biological therapy.<sup>1</sup> Despite advancements in therapy, CD can negatively impact long-term outcomes such as growth, pubertal development, and bone mineral density, leading to psychological and emotional effects, especially in adolescents.<sup>2-5</sup> Therefore, it is crucial to identify patients who will have a severe disease course and related risk variables already at diagnosis. Although several studies have attempted to pinpoint prognostic factors of

Received for publication: May 24, 2023. Editorial Decision: November 1, 2023

<sup>©</sup> The Author(s) 2023. Published by Oxford University Press on behalf of Crohn's & Colitis Foundation. All rights reserved. For permissions, please e-mail: journals.permissions@oup.com

#### **Key Messages**

#### What is already known?

 Pediatric Crohn's disease is characterized by heterogeneous clinical manifestations and an unpredictable disease course.

#### What is new here?

 About one-third of children with Crohn's disease experience a quiescent course during a 5-year follow-up, while half of them still have an active disease by the end of the follow-up. Early surgery is related to a mild disease course, while growth failure and lack of response to induction therapy are related to a worse prognosis.

#### How can this study help patient care?

 Identifying early prognostic risk factors for disease courses can help clinicians adjust therapeutic strategies based on early risk stratification.

disease course in both children and adults, in real-life practice each patient's disease course remains somewhat unpredictable.<sup>6-8</sup> This can make it challenging to decide on a tailored treatment strategy early at the diagnosis. In 2021, Wintjens et al<sup>9</sup> reported data from a cohort of 432 adult patients with CD, divided into 6 clusters of activity in a 10-year follow-up. About one-third of patients had a quiescent disease course, of whom 89% never received immunomodulators or biologics. Surgery at the diagnosis and higher age were positively associated with the quiescent course. Conversely, an ileocolonic location, smoking, and the need for steroids within 6 months from the diagnosis were related to a worse prognosis.9 Data on specific disease patterns in pediatric CD are lacking. Thus, this study aimed to define clusters of disease activity in pediatric CD based on clinical, laboratory, endoscopic, and therapeutic findings. As a secondary outcome, we sought to identify early prognostic risk factors for disease courses based on clinical and laboratory variables at diagnosis and in the first 6 months of follow-up.

## **Methods**

This is a retrospective, multicenter study involving pediatric patients affected with CD enrolled in the prospective SIGENP IBD (Italian Society of Pediatric Gastroenterology Hepatology and Nutrition Inflammatory Bowel Disease) registry, started on January 1, 2009. The study end date is considered as the date of the last visit before March 30, 2022. The methodology of the SIGENP IBD registry has been previously reported in detail.<sup>10</sup> Written consent was obtained from all patients and parents, and the study was approved by regional Ethics Committees, according to the rules of SIGENP IBD registry (Coordinating center approval, Sapienza University, Umberto I Hospital n. 4654). Eligible subjects included CD patients <18 years of age, with a 6-month follow-up during the first 5 years after diagnosis. All patients were enrolled at the diagnosis of CD. Patients with a shorter follow-up or incomplete data were excluded. According to Porto criteria, the diagnosis of IBD was based on clinical history, physical examination, endoscopic and histologic findings, and imaging.<sup>11</sup> Data collected for this study included demographic

features (age, sex), family history for IBD, disease location and behavior according to the Paris classification,<sup>12</sup> therapy at diagnosis, and follow-up. Disease activity was defined by the weighted Pediatric Crohn's Disease Activity Index (wPCDAI).<sup>13</sup> Laboratory data included erythrocyte sedimentation rate (ESR), C-reactive protein (CRP), platelet count, hemoglobin, albumin, and fecal calprotectin (FC), recorded at diagnosis and every 6 months throughout the follow-up. ESR was considered abnormal for values >20 mm/h, CRP >0.6 mg/dL, platelet count >400  $\times$  10<sup>9</sup>/L, hemoglobin <10 g/ dL, albumin <3.5 g/dL, and FC >250 µg/g. Endoscopic and/ or imaging evaluations (magnetic resonance enterography, ultrasonography) were recorded at diagnosis and follow-up, when available. The endoscopic activity was defined by the Simple Endoscopic Score for Crohn's Disease (SES-CD).<sup>14</sup> Mucosal healing was defined as SES-CD <3. When endoscopy was not available, FC <250 µg/g or Mucosal Inflammation Noninvasive Index (MINI) <8 were used as surrogate markers of mucosal healing. The MINI is a clinical-laboratory score based on 3 main items (frequency and characteristics of stool, FC, ESR, and CRP), with an overall score ranging from -3 to a maximum of 25 points.<sup>15</sup> A bowel wall thickness >3mm with signs of active inflammation on magnetic resonance enterography or ultrasonography (ie, the presence of an abnormal stratification, hypervascularization, mesenteric inflammatory fat, abscesses, or fistulas) was considered an active disease, regardless of the location.<sup>16</sup>

Treatment escalation was defined as the need for additional medical therapy based on a lack of sustained response/remission with the use of the previous maintenance drug: immunomodulators (azathioprine/6-mercaptopurine or methotrexate), biological therapies (anti-tumor necrosis factor  $\alpha$  [anti-TNF $\alpha$ ], vedolizumab, and ustekinumab), 5-aminosalicylic acid, or partial enteral nutrition.

Surgical procedures (bowel resection) were assessed at each time point. Disease progression was defined as a switch of CD phenotype from B1 to B2 or B3 or B2/B3 (12).

The previous variables were assessed on a 6-monthly basis during a 60-month follow-up. A disease flare was defined by the presence of 1 or more of these criteria at each semester: clinical activity (wPCDAI  $\geq$ 12.5 or MINI  $\geq$ 8) and/or SES-CD >3 or FC >250 µg/g and/or active disease at imaging methods.

## **Definition of Clusters**

The disease course was defined by the number of semesters with a flare during the first 5 years of follow-up. Then, based on the disease course, patients were classified into 1 of 5 previously identified disease activity clusters<sup>9</sup>: cluster A, active to remission (≥2 semesters of activity in the first 2 years, <2 semesters from 25-60 months); cluster B, remission to active (<2 semesters of activity in the first 2 years, ≥2 semesters from 25-60 months); cluster C, moderate–severe chronically active (≥1 semester of activity/year); cluster D, chronic intermittent (≥1 semester of activity every 2 years or an irregular chronic– intermittent, inactive–active–inactive, or vice versa pattern); and cluster E, quiescent (<2 semesters of activity on the entire 5-year follow-up).

#### Statistical Methods

All data were summarized and expressed as mean  $\pm$  SD or median and interquartile range (IQR) for continuous variables. Categorical data were expressed as frequencies

and percentages. Differences between groups, if any, were assessed by Student's t test. For qualitative variables, Fisher's correction chi-square statistical test was applied. The different clusters of patients in this population were identified based on previously published known patterns of disease activity in adults by considering either the presence or absence of disease recurrence for each 6-month follow-up.<sup>9</sup> The cumulative risk of surgical resection was calculated on the entire cohort of patients by assessing surgery-free survival using the Kaplan-Meier method. Univariate analysis was performed by simple logistic regression of variables detected at diagnosis and 6-month follow-up predicting quiescent course (cluster E), disease progression, and surgery. Multiple logistic regression was then performed using the significant variables in univariate. The odds ratio (OR), 95% confidence interval (CI), and P value were calculated in each analysis. A P value <.05 was considered statistically significant. GraphPad statistical software (GraphPad Instat 3.1 and Prism 8.4.2; GraphPad Software) was used to perform all statistical tests.

## Results

#### Demographic Data

During the inclusion period, 1019 children diagnosed with CD were enrolled in the registry. Of these, 687 were excluded (median age 13.1 years [IQR; 8.3-16.2 years]): 435 due to a follow-up <5 years, 189 due to lack of exhaustive data at each time point, and 63 due to a lack of each semester evaluation. Thus, 332 patients were enrolled in this study (142 [43%] females, median age 12 years [IQR, 8-14 years]). At diagnosis, 17 (5%) patients presented with severe disease activity, and 73 (22%), 192 (58%), and 50 (15%) with moderate, mild, and no clinical activity, respectively. Thirty-nine (12%) had an ileal location, 51 (15%) had a colonic disease, 144 (44%) had an ileocolonic disease, 21 (6%) had an isolated small bowel involvement, and 77 (23%) had a panenteric disease. Sixtyfive (20%) children presented with perianal involvement at the diagnosis. An inflammatory phenotype (B1) was present in 253 (76%) patients at the diagnosis, stricturing (B2) in 44 (13%), penetrating (B3) in 32 (10%), and B2/B3 in 3 (1%). The baseline characteristics of the population are shown in Table 1.

#### Clusters of Disease Activity and Clinical Outcomes

A total of 104 (31%) had active disease in the first 2 years, then substantial remission (cluster A), 43 (13%) were classified as cluster B (remission to active), 49 (15%) had a moderate-severe chronically active disease course (cluster C), and 31 (9%) were chronic-intermittent (cluster D) and 105 (32%) were quiescent (cluster E). Overall, 123 (37%) experienced a mostly active disease course at the end of follow-up. The disease patterns in the entire cohort are shown in Figure 1. No significant differences in age, location, and behavior at diagnosis were found. Cluster E (the quiescent pattern) was compared with all other clusters (Supplementary Table 1). Growth impairment at disease onset was significantly more frequent in clusters A and D compared with cluster E (P= .008 and P=.04, respectively), and persistent high ESR at 6 months from diagnosis was significantly related to cluster C than cluster E (P= .01). On the other hand, a significantly higher rate of patients with a quiescent disease course underwent

surgery at the diagnosis than in cluster A (P= .01), without any differences between other clusters.

In the univariate logistic regression analysis, a moderatesevere disease onset (wPCDAI >40), along with growth impairment at the diagnosis and persistently active disease at 6 months (ESR >20 mm/h and wPCDAI >40) requiring corticosteroids were inversely related to a quiescent disease course in the subsequent follow-up. Surgery at diagnosis was strongly related to the quiescent course (OR, 10.05; 95% CI, 3.05-25.22; P=0005). In a multivariable logistic regression model, clinically moderate–severe disease onset and the need for surgery at diagnosis maintained statistical significance (OR, 0.17; 95% CI, 0.06-0.43; P = .0005; and OR, 11.28; 95% CI, 1.44-25.00; P= .04, respectively; area under the curve [AUC], 0.76; 95% CI, 0.68-0.84; positive predictive value of 57% and negative predictive value [NPV] of 80%) (Table 2).

At maximum follow-up, 60 (18%) patients presented disease progression (10% in cluster A, 28% in B, 40% in C, 24% in D, and 8% in E), with a statistically significant difference between clusters C and E (P < .001). Fifty-three percent of patients progressed from inflammatory to stricturing disease, 33% to penetrating, and 14% to combined B2/B3. In multiple logistic regression analysis, perianal involvement at diagnosis and persistently active disease (ESR >20 mm/h and wPCDAI >40) correlated with disease progression at follow-up (AUC, 0.69; 95% CI, 0.59-0.79; NPV = 78%) (Table 3).

#### Medications

In the first year after diagnosis, the use of corticosteroids significantly decreased, while the use of biologics increased in the entire patient cohort. This trend remained stable over the medium and long term (Supplementary Figure 1). At the end of the follow-up, 139 (42%) patients were on biologic therapy, 93 (28%) were on immunomodulators, 67 (20%) were on 5-aminosalicylic acid, and 33 (10%) required systemic steroids. None of the patients were on partial enteral nutrition. Among those on biologics, 115 (35%) were receiving anti-TNF $\alpha$ , 15 (5%) ustekinumab, and 9 (3%) vedolizumab. No patients had discontinued medical treatment. At 5 years, there was a significant increase in the use of immunomodulators and biologics in clusters E and B, respectively (P= .0002 and P < .0001, respectively), compared with the others. No significant differences were found for other treatments according to disease clusters (Figure 2).

## Surgery

During the 5-year follow-up, 73 (22%) children underwent bowel resection. A total of 43% of these surgeries occurred within the first 6 months after diagnosis. Among the quiescent group (cluster E), 11 (11%) patients required surgery at diagnosis, with 9 more during the follow-up. A significantly higher proportion of patients in cluster C (33%) needed surgery than all other clusters (P=.04). Figure 3 shows the surgery-free survival rate in the entire cohort and the distribution by disease clusters. The logistic regression model showed that stricturing or penetrating phenotype at disease onset (OR, 3.04; 95% CI, 1.54-5.93; P=.001; OR, 9.24; 95% CI, 4.36-20.48; P < .001) along with perianal involvement (OR, 6.63; 95% CI, 3.65-12.19; P < .0001) and anti-TNF $\alpha$  use at diagnosis (OR, 3.15; 95% CI, 1.61-6.09; P=.0007) were significantly related to surgery risk at 5-year follow-up. Among these factors, only

Table 1. Baseline characteristics of 332 children with CD (N = 332).

Female	142 (43)
Age, y	12 (8-14)
A1a (<10 y at CD onset)	97 (29)
wPCDAI at diagnosis	$27 \pm 14$
<12.5 (remission)	50 (15)
12.5-40 (mild)	192 (58)
40-57.5 (moderate)	73 (22)
$\geq$ 57.5 (severe)	17 (5)
CRP, mg/dL	$2.3 \pm 0.55$
FC, µg/g	$760 \pm 518$
ESR, mm/h	$42 \pm 20$
CD location at diagnosis	
L1	39 (12)
L2	51 (15)
L3	144 (44)
L4a	77 (23)
L4b	21 (6)
Perianal disease at diagnosis	65 (20)
Behavior at diagnosis	
B1	253 (76)
B2	44 (13)
B3	32 (10)
B2/B3	3 (1)
Therapy at diagnosis	
CS	153 (46)
5-ASA	113 (34)
IFX	37 (11)
ADA	6 (2)
AZA	154 (46)
MTX	6 (2)
EEN	132 (40)

Values are n (%), median (interquartile range), or mean ± SD. Abbreviations: 5-ASA, 5-aminosalicylic acid; ADA, adalimumab; AZA, azathioprine; CD, Crohn's disease; CRP, C-reactive protein; CS, corticosteroids; EEN, exclusive enteral nutrition; ESR, erythrocyte sedimentation rate; FC, fecal calprotectin; IFX, infliximab; MTX, methotrexate; wPCDAI, weighted Pediatric Crohn's Disease Activity Index.

disease phenotype and perianal involvement maintained statistical significance on multiple regression analysis (AUC, 0.76; 95% CI, 0.69-0.83; NPV 82%; positive predictive value 61%) (Table 4). Age, disease location, growth impairment, need for systemic steroids, and exclusive enteral nutrition did not significantly influence the risk of surgery.

## Discussion

To the best of our knowledge, this is the first cohort study identifying patterns of clinical activity and early risk factors for disease courses in pediatric CD.

Our data show that around one-third of pediatric patients experience a mild long-term disease course after an initial activity phase. However, 37% have an active disease during the entire follow-up, either persistently active, chronically intermittent, or persistently active after an initial remission. Our data are consistent with previously published data in adults over 10 years of follow-up,<sup>9,17-19</sup> although we found a

higher percentage of children with a persistently active disease course. Solberg et al<sup>19</sup> recently reported similar results in a cohort of adult patients with CD. In this study, the authors identified 4 disease patterns by submitting questionnaires to enrolled patients. Forty percent of them reported a quiescent or mild disease course at follow-up. This finding is higher than our cohort, although the method based on a patient survey could have determined biases in defining active disease. It is worth noting that comparing pediatric and adult outcomes does not account for significant differences between the 2 groups. Chronic active disease can have a negative impact on a child's growth, pubertal development, and psychological well-being, which may be irreversible.<sup>20-24</sup> Moreover, the need for continuous medical therapies to control a persistently active disease is a significant concern. In our study, we found that most children with a quiescent disease course were on immunomodulators or biologics. Interestingly, in a similar study by Wintjens et al,<sup>9</sup> about 90% of adult patients with the same disease course never used immunosuppressives or biologics.

Additionally, our study revealed that immunomodulators are widely used to sustain remission in pediatric CD. This finding is consistent with other data showing that this class of drugs is commonly used as the maintenance therapy in pediatric CD and UC, even in the biologic era.<sup>25-30</sup>

As a secondary outcome, we analyzed several variables at diagnosis and in the first 6 months after diagnosis to identify predictive factors of the disease course.

Growth impairment at diagnosis and persistently active disease 6 months after diagnosis were associated with a more aggressive disease course. These results are in keeping with previous data by Ziv-Baran et al.<sup>31</sup> In their study, an early response to remission-inducing therapies correlated with a lower risk of disease recurrence and complications in the medium term, much more than the disease severity at the diagnosis. A more recent pediatric study also supports this finding, showing that clinical and laboratory response to exclusive enteral nutrition as induction therapy in CD is related to favorable long-term outcomes and a lower risk of recurrence.<sup>32</sup>

We also assessed the early risk factors of disease progression from noncomplicated to complicated disease behaviors in the entire cohort of patients. Perianal disease at diagnosis and persistently active disease 6 months after diagnosis were related to a high risk of disease progression. This is consistent with data highlighting the role of perianal disease at onset as a risk factor for disease progression.<sup>33-35</sup> Recently, a systematic review analyzing prognostic risk factors in CD identified small bowel disease location, elevated levels of anti-*Saccharomyces cerevisiae* antibodies, and polymorphisms in the NOD2/CARD15 genes as prognostic factors of a worse prognosis in pediatric CD.<sup>6</sup> Our research did not include an autoantibodies profile and genetic evaluation because they were unavailable in most patients.

We found a 5-year surgical rate of 22%. These data align with previously published reports ranging from 13% to 35%.<sup>36,37</sup> It is worth noting that a relatively high number of patients underwent surgery within the first 6 months of their diagnosis. We also found that early surgery was related to a quiescent course at follow-up, a finding that was also reported by Wintjens et al<sup>9</sup> in adults with CD. It is also interesting that the randomized, prospective, open-label LIR!C (Laparoscopic ileocaecal resection versus infliximab for



Figure 1. Clusters of disease activity in a cohort of 332 children with Crohn's disease.

terminal ileitis in Crohn's disease) trial, conducted in 143 adults with limited ileocecal disease and who have failed conventional therapy, demonstrated that ileocecal resection could be considered a valid alternative to infliximab therapy in terms of long-term quality of life.<sup>38</sup> The significance of early surgery as a predictor of a mild disease course is noteworthy because it suggests that in certain pediatric CD patients, a limited bowel resection early on after diagnosis may improve their long-term outcomes and potentially reduce the impact of medical therapies.

In line with other data, we found the inflammatory phenotype to be inversely correlated with the surgical risk, while complicated phenotypes along with perianal disease at diagnosis were related to increased surgical risk.<sup>39-42</sup> We did not find any significant difference in age and disease location. This finding contradicts a recent meta-analysis that found that older children with an isolated ileal location are at a higher risk of surgery.<sup>6</sup>

Interestingly, in the univariate analysis, anti-TNF $\alpha$  use at diagnosis and at 6 months significantly correlated with the surgical risk, although this result was not maintained in the multivariate analysis. This result is somewhat in line with the results of the RISK study (Risk Stratification and Identification of Immunogenetic and Microbial Markers of Rapid Disease Progression in Children with Crohn's Disease), showing that early anti-TNF therapy is not associated with a reduction of fibrostenotic complications (while it reduces the risk of penetrating phenotypes).<sup>43</sup> Table 2. Univariate and multivariate analysis of early variables related to a quiescent disease course (cluster E).

	Univariate			Multivariate		
	OR	95% CI	Р	OR	95% CI	Р
wPCDAI >40 at diagnosis	0.15	0.06-0.31	<.0001ª	0.17	0.06-0.43	.0005ª
L2	1.12	0.61-1.99	.68			
G1	0.48	0.27-0.81	.007ª	0.93	0.39-2.25	.87
B1	0.69	0.38-1.26	.21			
Age <6 y	1.45	0.68-2.96	.32			
Age <10 y	1.37	0.82-2.29	.23			
EEN at diagnosis	1.26	0.78-2.04	.34			
ESR >20 mm/h at diagnosis	1.11	0.53-2.45	.79			
Perianal disease	1.02	0.55-1.83	.95			
CS at diagnosis	1.03	0.64-1.54	.9			
Surgery at diagnosis	10.05	3.05-25.22	.0005ª	11.28	1.44-25.00	.04ª
wPCDAI >40 at 6 mo	0.21	0.06-0.54	.004ª	0.39	0.08-1.42	.18
ESR >20 mm/h at 6 mo	0.42	0.22-0.79	.007ª	0.55	0.25-1.19	.13
CS at 6 mo	0.36	0.16-0.71	.005ª	0.03	0.35-2.06	.89
Surgery at 6 mo	0.7	0.19-2.06	.55			

Model characteristics were the following: area under the curve, 0.76; 95% CI, 0.68-0.84;  $R^2 = 0.17$ ; positive predictive value = 57%; negative predictive value = 80%.

Abbreviations: CI, confidence interval; CS, corticosteroids; EEN, exclusive enteral nutrition; ESR, erythrocyte sedimentation rate; OR, odds ratio; wPCDAI, weighted Pediatric Crohn's Disease Activity Index. <sup>a</sup>p<.05.

Table 3. Univariate and multivariate analysis of early variables related to disease progression in all patients.

	Univariate			Multivariate		
	OR	95% CI	Р	OR	95% CI	Р
wPCDAI >40 at diagnosis	1.94	0.98-3.76	.05ª			
L2	0.78	0.39-1.46	.45			
G1	1.24	0.73-2.08	.42			
Age <6 y	0.44	0.07-1.60	.27			
Age <10 y	0.82	0.46-1.43	.5			
Age 10-17 y	1.18	0.68-2.09	.56			
ESR >20 mm/h at diagnosis	1.47	0.64-3.80	.39			
Perianal disease	1.84	1.01-3.30	.04ª	2.55	1.05-6.07	.03
EEN at diagnosis	1.16	0.69-1.93	.56			
CS at diagnosis	0.83	0.49-1.37	.46			
Anti-TNFα at diagnosis	1.33	0.64-2.63	.41			
wPCDAI >40 at 6 mo	3.21	1.28-7.96	.01ª	3.85	1.20-12.55	.02
ESR >20 mm/h at 6 mo	2.18	1.06-4.68	.04ª	1.55	0.69-3.62	.29
CS at 6 mo	0.78	0.39-1.50	.48			
Anti-TNFα at 6 mo	1.64	0.91-2.89	.09ª			

Model characteristics were the following: area under the curve, 0.69; 95% CI, 0.59-0.79;  $R^2 = 0.08$ ; positive predictive value = 50%; negative predictive value = 77.85%.

Abbreviations: CI, confidence interval; CS, corticosteroids; EEN, exclusive enteral nutrition; ESR, erythrocyte sedimentation rate; OR, odds ratio; TNF $\alpha$ , tumor necrosis factor  $\alpha$ ; wPCDAI, weighted Pediatric Crohn's Disease Activity Index. <sup>a</sup>p<.05.

One may speculate that patients with more severe disease at diagnosis who start anti-TNF $\alpha$  upfront may have a higher inflammatory burden and complicated diseases commonly requiring surgery during the disease course. The role of biological therapy as a risk factor for surgery remains unclear, with some studies suggesting a correlation and others showing opposite results.<sup>42,44-46</sup>

Our study has some notable strengths. First, we included a large cohort of pediatric patients and we used stringent criteria to evaluate the disease course. Second, we applied a



CS, corticosteroids; IM, immunomodulator; BIO, biologics; 5-ASA, 5-aminosalicylic acid; EEN, Exclusive Enteral Nutrition; PEN, Partial Enteral Nutrition.

Figure 2. Therapeutic strategies in each disease cluster at diagnosis (left) and the end of the 5-year follow-up (right).



Figure 3. Surgery-free survival curve (left) and percentage and numbers of patients who underwent surgery for each disease cluster (right).

	Univariate			Multivariate		
	OR	95% CI	Р	OR	95% CI	Р
wPCDAI >40 at diagnosis	1.47	0.75-2.79	.25			
L2	1.12	0.58-2.07	.73			
L3	0.86	0.51-1.47	.57			
G1	0.85	0.48-1.47	.55			
B1	0.19	0.10-0.35	.001ª	0.85	0.29-2.48	.77
B2	3.04	1.54-5.93	.001ª	3.68	1.21-10.89	.01ª
B3	9.24	4.36-20.48	<.001 <sup>a</sup>	4.28	1.38-13.78	.012ª
EEN at diagnosis	1.07	0.63-1.81	.79			
ESR >20 mm/h at diagnosis	1.08	0.48-2.67	.86			
Perianal disease	6.63	3.65-12.19	<.0001ª	2.99	1.29-6.77	.008ª
CS at diagnosis	0.72	0.42-1.21	.22			
Anti-TNFα at diagnosis	3.15	1.61-6.09	.0007ª	1.71	0.69-4.17	.24
wPCDAI >40 at 6 mo	0.96	0.38-2.19	.93			
ESR >20 mm/h at 6 mo	1.95	0.99-3.90	.053			
CS at 6 mo	0.47	0.21-0.97	.054			
Anti-TNF $\alpha$ at 6 mo	1.95	1.10-3.42	.02ª	1.06	0.49-2.20	.89

**Table 4.** Logistic regression of early predictive variables related to surgery risk.

Model characteristics were the following: area under the curve, 0.76; 95% CI, 0.69-0.83;  $R^2 = 0.20$ ; positive predictive value = 61%; negative predictive value = 82%.

Abbreviations: CI, confidence interval; CS, corticosteroids; EEN, exclusive enteral nutrition; ESR, erythrocyte sedimentation rate; OR, odds ratio; TNF $\alpha$ , tumor necrosis factor  $\alpha$ ; wPCDAI, weighted Pediatric Crohn's Disease Activity Index. <sup>a</sup>p<.05.

7

rigorous definition of disease activity, which was measured using clinical, laboratory, and multiparametric scores as well as endoscopic and/or imaging data. Last, we only included patients with complete data for each semester, which allowed us to accurately define disease recurrence during follow-up and cluster classification.

However, we are also aware of the weaknesses of our study. First, we collected retrospective data, which were, however, recorded prospectively in the SIGENP national registry for pediatric IBD. Second, due to our inclusion criteria and the need for strict follow-up, a significant proportion of patients from the entire cohort were excluded, which could increase selection bias. Nonetheless, we analyzed the portion of the excluded patients due to incomplete assessments and found no significant differences in the main outcomes analyzed. Therefore, we believe that our cohort reflects the real-life course of CD in children.

# Conclusions

During a 5-year follow-up, about one-third of children with CD experience a quiescent course. However, nearly half of them still have an active disease by the end of the follow-up, with a persistently active or chronic-intermittent course or active after an initial remission. The severity of the disease at onset, growth impairment, and no response to induction therapy are associated with a more aggressive disease course. Conversely, early surgery is linked to a milder disease accurse. The 5-year surgical rate is 22%, with most surgeries occurring in the first 2 years of the disease. Complicated behaviors and the presence of perianal disease at the diagnosis increase the risk of bowel resection. Clinicians can adjust their therapeutic strategies based on an early risk stratification considering all these criteria to improve long-term disease outcomes.

# Supplementary data

Supplementary data is available at *Inflammatory Bowel Diseases* online.

# Acknowledgments

The authors wish to thank the following member of the SIGENP IBD working group for their helpful discussion and review process of this study: M. Baldi, C. Banzato, G. Barera, G. Castellucci, M. Corpino, R. Cozzali, C. De Giacomo, P. Diaferia, D. Dilillo, E. Felici, M.T. Illiceto, D. Knafelz, L. Norsa, P.M. Pavanello, A. Ravelli, V. Romagnoli, C. Romano, and S. Salvatore.

# **Author Contribution**

M.A. designed the study, wrote the manuscript, and approved the version to be published. M.D., S.R., and M.R. contributed to design and to write the manuscript. E.P., P.L., S.A., and P.A. contributed to design and write the manuscript. E.M., M.M., and G.Z. collected the data and made the statistical analysis. M.B. and M.C. contributed to collect the data and revised the article.

# Funding

No funding has been received for this work.

# **Conflicts of Interest**

None declared. All the authors declare no financial relationships with a commercial entity producing healthrelated products and or services related to this article. No honorarium, grant, or other form of payment was given to anyone to write and to produce the manuscript.

## References

- 1. Van Rheenen PF, Aloi M, Assa A, et al. The medical management of paediatric Crohn's disease: an ECCO-ESPGHAN guideline update. *J Crohns Colitis*. 2020;15(2):171-194.
- 2. Brain CE, Savage MO. Growth and puberty in chronic inflammatory bowel disease. *Baillieres Clin Gastroenterol.* 1994;8(1):83-100.
- Mittermaier C, Dejaco C, Waldhoer T, et al. Impact of depressive mood on relapse in patients with inflammatory bowel disease: a prospective 18-month follow-up study. *Psychosom Med.* 2004;66(1):79-84.
- Walther F, Fusch C, Radke M, Beckert S, Findeisen A. Osteoporosis in pediatric patients suffering from chronic inflammatory bowel disease with and without steroid treatment. J Pediatr Gastroenterol Nutr. 2006;43(1):42-51.
- Schmidt S, Mellström D, Norjavaara E, Sundh SV, Saalman R. Low bone mineral density in children and adolescents with inflammatory bowel disease: a population-based study from Western Sweden. *Inflamm Bowel Dis.* 2009;15(12):1844-1850.
- Ricciuto A, Aardoom M, Orlanski-Meyer E, et al.; Pediatric Inflammatory Bowel Disease–Ahead Steering Committee. Predicting outcomes in pediatric Crohn's disease for management optimization: systematic review and consensus statements from the pediatric inflammatory bowel disease-ahead program. *Gastroenterology*. 2021;160(1):403-436.e26.
- Klamt J, De Laffolie J, Wirthgen E, Stricker S, Däbritz J; CEDATA-GPGE study group. Predicting complications in pediatric Crohn's disease patients followed in CEDATA-GPGE registry. *Front Pediatr.* 2023;11:1043067.
- Karoui S, Serghini M, Dachraoui A, Boubaker J, Filali A. Prognostic factors in Crohn's disease: a systematic review. *Tunis Med*. 2013;91(4):230-233.
- 9. Wintjens D, Bergey F, Saccenti E, et al. Disease activity patterns of Crohn's disease in the first 10 years after diagnosis in the population-based IBD South Limburg Cohort. J Crohns Colitis. 2021;15(3):391-400.
- Aloi M, Lionetti P, Barabino A, et al.; SIGENP IBD Group. Phenotype and disease course of early-onset pediatric inflammatory bowel disease. *Inflamm Bowel Dis.* 2014;20(4):597-605.
- Levine A, Koletzko S, Turner D, et al.; European Society of Pediatric Gastroenterology, Hepatology, and Nutrition. ESPGHAN revised Porto criteria for the diagnosis of inflammatory bowel disease in children and adolescents. *J Pediatr Gastroenterol Nutr.* 2014;58(6):795-806.
- 12. Levine A, Griffiths A, Markowitz J, et al. Pediatric modification of the Montreal classification for inflammatory bowel disease: the Paris classification. *Inflamm Bowel Dis.* 2011;17(6):1314-1321.
- 13. Turner D, Griffiths AM, Walters TD, et al. Mathematical weighting of the pediatric Crohn's disease activity index (PCDAI) and comparison with its other short versions. *Inflamm Bowel Dis.* 2012;18(1):55-62.
- 14. Daperno M, D'Haens G, Van Assche G, et al. Development and validation of a new, simplified endoscopic activity score for Crohn's disease: the SES-CD. *Gastrointest Endosc.* 2004;60(4):505-512.
- 15. Cozijnsen MA, Ben Shoham A, Kang B, et al. Development and validation of the mucosal inflammation noninvasive index for pediatric Crohn's disease. *Clin Gastroenterol Hepatol.* 2020;18(1):133-140.e1.
- 16. Ma L, Li W, Zhuang N, et al. Comparison of transmural healing and mucosal healing as predictors of positive long-term

outcomes in Crohn's disease. *Therap Adv Gastroenterol.* 2021;14:17562848211016259.

- Henriksen M, Jahnsen J, Lygren I, et al.; IBSEN Study Group. Ulcerative colitis and clinical course: results of a 5-year populationbased follow-up study (the IBSEN study). *Inflamm Bowel Dis*. 2006;12(7):543-550.
- Solberg IC, Lygren I, Jahnsen J, et al.; IBSEN Study Group. Clinical course during the first 10 years of ulcerative colitis: results from a population-based inception cohort (IBSEN Study). *Scand J Gastroenterol.* 2009;44(4):431-440.
- Solberg IC, Vatn MH, Høie O, et al.; IBSEN Study Group. Clinical course in Crohn's disease: results of a Norwegian populationbased ten-year follow-up study. *Clin Gastroenterol Hepatol.* 2007;5(12):1430-1438.
- Sauer CG, Kugathasan S. Pediatric inflammatory bowel disease: highlighting pediatric differences in IBD. *Gastroenterol Clin North Am.* 2009;38(4):611-628.
- 21. Szigethy E, Levy-Warren A, Whitton S, et al. Depressive symptoms and inflammatory bowel disease in children and adolescents: a cross-sectional study. *J Pediatr Gastroenterol Nutr.* 2004;39(4):395-403.
- 22. Jin HY, Lim JS, Lee Y, et al. Growth, puberty, and bone health in children and adolescents with inflammatory bowel disease. *BMC Pediatr.* 2021;21(1):35.
- Wong K, Isaac DM, Wine E. Growth delay in inflammatory bowel diseases: significance, causes, and management. *Dig Dis Sci.* 2021;66(4):954-964.
- 24. Claar RL, van Tilburg MAL, Abdullah B, et al. Psychological distress and quality of life in pediatric Crohn's disease: impact of pain and disease state. *J Pediatr Gastroenterol Nutr.* 2017;65(4):420-424.
- 25. Jongsma MME, Aardoom MA, Cozijnsen MA, et al. First-line treatment with infliximab versus conventional treatment in children with newly diagnosed moderate-to-severe Crohn's disease: an open-label multicentre randomised controlled trial. *Gut.* 2022;71(1):34-42.
- Magro F, Cordeiro G, Dias AM, Estevinho MM. Inflammatory bowel disease - non-biological treatment. *Pharmacol Res.* 2020;160:105075.
- Kapur S, Hanauer SB. The evolving role of thiopurines in inflammatory bowel disease. *Curr Treat Options Gastroenterol*. 2019;17(3):420-433.
- Baldwin K, Goldfarb A, Brimacombe M, Hopkins D, Hyams JS. Methotrexate for primary maintenance therapy in mild-tomoderate Crohn disease in children. *J Pediatr Gastroenterol Nutr.* 2022;75(3):320-324.
- El-Matary W, Carroll MW, Debruyn JC, et al. Discontinuation of immunosuppressive medications in children with inflammatory bowel disease on combination therapy. J Pediatr Gastroenterol Nutr. 2020;71(6):740-743.
- Luber RP, Dawson L, Munari S, et al. Thiopurines and their optimization during infliximab induction and maintenance: a retrospective study in Crohn's disease. J Gastroenterol Hepatol. 2021;36(4):990-998.
- 31. Ziv-Baran T, Hussey S, Sladek M, et al. Response to treatment is more important than disease severity at diagnosis for prediction of early relapse in new-onset paediatric Crohn's disease. *Aliment Pharmacol Ther.* 2018;48(11-12):1242-1250.

- 32. Herman R, Dumnicka P, Pieczarkowski S, Fyderek K. Induction of remission in pediatric Crohn's disease patients assessed by the mucosal inflammation noninvasive index. *J Clin Med.* 2021;10(23):5613.
- 33. Atia O, Focht G, Lujan R, et al. Perianal Crohn disease is more common in children and is associated with complicated disease course despite higher utilization of biologics: a populationbased study from the epidemiology group of the Israeli IBD Research Nucleus (epiIIRN). J Pediatr Gastroenterol Nutr. 2022;74(6):788-793.
- Assa A, Amitai M, Greer ML, et al.; ImageKids Study Group. Perianal pediatric Crohn disease is associated with a distinct phenotype and greater inflammatory burden. *J Pediatr Gastroenterol Nutr.* 2017;65(3):293-298.
- 35. Fumery M, Pariente B, Sarter H, et al.; Epimad Group. Long-term outcome of pediatric-onset Crohn's disease: a population-based cohort study. *Dig Liver Dis*. 2019;51(4):496-502.
- Rinawi F, Assa A, Hartman C, et al. Incidence of bowel surgery and associated risk factors in pediatric-onset Crohn's disease. *Inflamm Bowel Dis.* 2016;22(12):2917-2923.
- Vernier-Massouille G, Balde M, Salleron J, et al. Natural history of pediatric Crohn's disease: a population-based cohort study. *Gastroenterology*. 2008;135(4):1106-1113.
- Ponsioen CY, De Groof EJ, Eshuis EJ, et al.; LIR!C study group. Laparoscopic ileocaecal resection versus infliximab for terminal ileitis in Crohn's disease: a randomised controlled, open-label, multicentre trial. *Lancet Gastroenterol Hepatol*. 2017;2(11):785-792.
- Schaefer ME, Machan JT, Kawatu D, et al. Factors that determine risk for surgery in pediatric patients with Crohn's disease. *Clin Gastroenterol Hepatol.* 2010;8(9):789-794.
- 40. Kurowski JA, Milinovich A, Ji X, et al. Differences in biologic utilization and surgery rates in pediatric and adult Crohn's disease: results from a large electronic medical record-derived co-hort. *Inflamm Bowel Dis.* 2021;27(7):1035-1044.
- 41. Rumenova Shentova-Eneva R, Kofinova D, et al. Risk factors for surgery in pediatric patients with Crohn's disease. *Med Princ Pract.* 2022;31(2):195-200.
- Kerur B, Machan JT, Shapiro JM, et al. Biologics delay progression of Crohn's disease, but not early surgery, in children. *Clin Gastroenterol Hepatol.* 2018;16(9):1467-1473.
- Kugathasan S, Denson LA, Walters TD, et al. Prediction of complicated disease course for children newly diagnosed with Crohn's disease: a multicentre inception cohort study. *Lancet*. 2017;389(10080):1710-1718.
- 44. Ley D, Leroyer A, Dupont C, et al.; Epimad Group. New therapeutic strategies have changed the natural history of pediatric Crohn's disease: a two-decade population-based study. *Clin Gastroenterol Hepatol.* 2022;20(11):2588-2597.e1.
- 45. Woo MH, Cho YH, Sohn MJ, et al. Use of anti-TNF alpha blockers can reduce operation rate and lead to growth gain in pediatric Crohn's disease. *Pediatr Gastroenterol Hepatol Nutr.* 2019;22(4):358-368.
- 46. Murthy SK, Begum J, Benchimol EI, et al. Introduction of anti-TNF therapy has not yielded expected declines in hospitalisation and intestinal resection rates in inflammatory bowel diseases: a population-based interrupted time series study. *Gut.* 2020;69(2):274-282.